Frequent oil baths and skin barrier during infancy in the PreventADALL study

Eva Maria Rehbinder,^{1,2} Sabina Wärnberg Gerdin,^{3,4} Angela Hoyer,^{3,4} Maria Bradley,⁵ Karin C. Lødrup Carlsen,^{2,6} Berit Granum,⁷ Gunilla Hedlin,^{3,4} Christine Monceyron Jonassen,^{8,9} Marissa Leblanc,^{10,11} Björn Nordlund,^{3,4} Knut Rudi,¹² Håvard O. Skjerven,^{2,6} Anne Cathrine Staff,^{2,13} Riyas Vettukattil^{2,6} and Cilla Söderhäll^{3,4}

¹Department of Dermatology, Oslo University Hospital, Oslo, Norway

²Faculty of Medicine, Institute of Clinical Medicine, University of Oslo, Oslo, Norway

³Department of Women's and Children's Health, Karolinska Institutet, Stockholm, Sweden

⁴Astrid Lindgren Children's Hospital, Karolinska University Hospital, Stockholm, Sweden

⁵Dermatology Unit, Department of Medicine Solna, Karolinska Institutet and Karolinska University Hospital, Stockholm, Sweden

⁶Divisions of Paediatric and Adolescent Medicine, Oslo University Hospital, Oslo, Norway

⁷Department of Chemical Toxicology, Norwegian Institute of Public Health, Oslo, Norway

⁸Center for Laboratory Medicine, Østfold Hospital Trust, Kalnes, Norway

⁹Department of Virology, Norwegian Institute of Public Health, Oslo, Norway

¹⁰Oslo Centre for Biostatistics and Epidemiology; University of Oslo, Oslo, Norway

¹¹Norwegian Institute of Public Health, Oslo, Norway

¹²Department of Chemistry, Biotechnology and Food Science, Norwegian University of Life Sciences, Ås, Norway

¹³Divisions of Obstetrics and Gynaecology; Oslo University Hospital, Oslo, Norway

Eva Maria Rehbinder and Sabina Wärnberg Gerdin share first authorship.

Correspondence: Eva Maria Rehbinder. Email: e.m.rehbinder@ous-research.no

Linked Article: Danby Br J Dermatol 2024; 191:6–7.

Abstract

Background In the general population randomized controlled trial PreventADALL, frequent emollient bath additives from 2 weeks of age did not prevent atopic dermatitis, while the effect on skin barrier function throughout infancy is not established.

Objectives The primary aim of this exploratory substudy was to assess the effect of mineral-based oil baths on transepidermal water loss (TEWL) and dry skin through infancy, and secondarily to explore if *filaggrin* (*FLG*) mutations modified the effect.

Methods Overall, 2153 infants were included and randomized to either the 'Skin intervention' (SI) group (n=995) (oil bath 4 times weekly from 2 weeks through 8 months) or 'No skin intervention' (NSI) group (n=1158), with TEWL measurements at 3, 6 and/or 12 months of age. Information on *FLG* mutation status was available for 1683 of these infants. Effects of the skin intervention on TEWL and dry skin through infancy were assessed by mixed-effects regression modelling. Background characteristics and protocol adherence were collected from electronic questionnaires, birth records and weekly diaries.

Results The TEWL (95% confidence interval) was on average 0.42 g m⁻² h⁻¹ (0.13–0.70, P=0.004) higher in the SI group compared with the NSI group through the first year of life, with significantly higher levels at 3 months [8.6 (8.3–9.0) vs. 7.6 (7.3–7.9)], but similar at 6 and 12 months. Dry skin was observed significantly more often in the NSI group compared with the SI group at 3 months (59% vs. 51%) and at 6 months of age (63% vs. 53%), while at 12 months of age, the difference was no longer significant. At 3 months, the TEWL of *FLG* mutation carriers was similar to the TEWL in the SI group. No interaction between SI and *FLG* mutation was found in the first year of life.

Conclusions Infants given frequent oil baths from 2 weeks of age had reduced skin barrier function through infancy compared with controls, largely attributed to higher TEWL at 3 months of age, while the skin at 3 and 6 months appeared less dry in infants subjected to the skin intervention.

Lay summary

Atopic dermatitis (AD) affects approximately 20% of children in industrialized countries. AD causes dry, itchy skin and can increase the chance of infections.

This study was a substudy of the large Scandinavian PreventADALL trial, including 2394 infants, recruited from the general population between 2014 and 2016. Children in this trial were allocated randomly to receive either a skin intervention, food intervention, combined

Accepted: 23 February 2024

© The Author(s) 2024. Published by Oxford University Press on behalf of British Association of Dermatologists. This is an Open Access article distributed under the terms of the Creative Commons Attribution License (https://creativecommons.org/licenses/by/4.0/), which permits unrestricted reuse, distribution, and reproduction in any medium, provided the original work is properly cited. intervention, or no intervention. Children were examined at 3, 6 and 12 months of age. The examinations involved an investigation of the skin, to evaluate dry skin and skin barrier function by transepidermal water loss (TEWL) in the outer layers of the skin (higher TEWL suggests decreased skin barrier function). The skin intervention consisted of oil baths at least 4 times per week from 2 weeks of age through 8 months of age, and have previously not been shown to prevent AD by 1 and 3 years of age. We aimed to investigate whether frequent oil baths had any effect on TEWL and dry skin.

We found that the skin intervention increased TEWL in the first year of life, especially at 3 months of age. Dry skin was less common in the skin intervention groups compared with the groups with no skin intervention. Infants with mutations in the gene coding for a skin barrier protein, called *filaggrin*, were associated with increased TEWL; however, in the skin intervention group, TEWL was similar among the infants with or without *filaggrin* mutations. Our findings suggest that oil baths several times per week from early infancy transiently decreases skin barrier function.

What is already known about this topic?

• Mineral-based oil baths several times per week from early infancy do not prevent atopic dermatitis.

What does this study add?

• New insights on the negative effect of paraffinum liquidum oil baths during early infancy on skin barrier function, with or without FLG mutations.

Atopic dermatitis (AD) affects around 20% of children in industrialized countries¹ and is characterized by dry, itchy and inflamed skin susceptible to infections.² The skin barrier function is usually impaired, leading to increased transepidermal water loss (TEWL) in both lesional and nonlesional atopic skin.^{3–6} Dry skin is associated with increased TEWL in children,^{5,7} including 3-month-old infants without eczematous lesions, as recently shown in the Preventing Atopic Dermatitis and ALLergies in children (PreventADALL) study.⁵ Both dry skin⁸ and increased TEWL⁹ has been found to precede AD, and TEWL also correlates with the severity of AD as well as response to treatment.^{10,11}

Filaggrin (*FLG*) encodes a skin barrier protein, which is essential for the epidermal differentiation, including the structure and function of the stratum corneum.¹² Loss-offunction mutations in the *FLG* gene impair the skin barrier function, and is therefore a major risk factor for the development of AD.¹³ Around 20–50% of patients with AD carry at least one *FLG* mutation,¹⁴ compared with 10% in the general population.¹⁵

The first-line AD treatment consists of improving the impaired skin barrier by moisturization.² Although it is well established that leave-on emollients in AD treatment reduce the severity and prevent flares,¹⁶⁻¹⁸ the effect of emollient bath additives in AD treatment is unclear.^{16,18-20} Also, the effects of emollients on TEWL are inconsistent; however, they are somewhat favourable for emollients containing humectants, such as urea or ammonium lactate, and when used to treat dry or atopic skin.^{16,18,20-26}

Although important in AD treatment, a meta-analysis including 11 randomized controlled trials (RCTs) found that regular emollients from early infancy failed to reduce the risk of AD development in early childhood.¹⁸ The two largest RCTs included in the meta-analysis with negative results using paraffinum liquidum-based emollients were our Scandinavian general population PreventADALL study including 2394 mother-child pairs using oil bath and emollient facial cream²⁷ and the British Barrier Enhancement for

Eczema Prevention (BEEP) study with 1394 high-risk infants using leave-on emollients.²⁸

In order to better understand why the skin intervention in the PreventADALL study did not prevent AD, we aimed primarily to assess the effect of the skin intervention on TEWL and dry skin through infancy, and secondarily to explore if the potential effects were influenced by carrying *FLG* mutations.

Materials and methods

Study design

The present study is an exploratory subanalysis of the PreventADALL study, a Scandinavian multicentre RCT and prospective birth cohort,²⁹ enrolling 2697 pregnant women from Oslo and Østfold (Norway) and Stockholm (Sweden), and collecting baseline characteristics through electronic questionnaire.

Study population

All 2153 of 2394 randomized infants, with TEWL measured on at least one of the clinical follow-up investigations at 3, 6 and/or 12 months of age were included in the primary analyses, while the secondary aim analysed the 1683 infants who also had information on *FLG* mutation status (Figure 1).

Randomization and interventions

Randomization was done in a 1 : 1 : 1 ratio into four groups: no intervention, skin intervention, food intervention or combined interventions. The skin intervention²⁷ consisted of baths for 5–10 min with added emulsified oil (0.5 dL bath oil per 8 L water) and daily facial cream (Ceridal[®]) on 4–7 days weekly from week 2 through 8 months of age. The bath oil was produced specifically for the PreventADALL trial by Pharmatech (Østfold, Norway), consisting of paraffinum liquidum and



Figure 1 Study population of the present study, from the Preventing Atopic Dermatitis and Allergies in children (PreventADALL) mother–child birth cohort, where 2701 pregnancies were included.

trilaureth-4-phosphate only. The use of emollients was recorded in weekly electronic diaries from 2 weeks to 26 weeks of age, and thereafter in 3-monthly electronic questionnaires. Full-skin intervention adherence was defined as oil baths and facial cream for an average of at least 3.5 days per week for at least 16 of 25 weeks, whereas partial adherence to the skin intervention is based on the average number of oil baths only in four categories: \geq 3.5, 2.5–3.4, 1.5–2.4 and 0.5–1.4 days per week.²⁷ The food intervention^{27,30} consisted of introducing four foods as small tastes (peanut, cow's milk, wheat and egg) between 12 and 16 weeks of age and maintaining in the regular diet to at least 6 months of age.

Transepidermal water loss

Skin barrier function was reported as TEWL, which is mean TEWL (g m⁻² h⁻¹) of three performed measurements at each time-point on the left lateral upper arm using an open-chamber DermaLab USB (Cortex, Hadsund, Denmark), noting ambient temperature (room temperature between 20 °C and 25 °C, with windows and doors shut) and humidity, in line with international recommendations³¹ and as described in our previous publications.^{5,8,32}

Dry skin

Dry skin was characterized by the presence of skin scaling and roughness both by visual inspection and palpation in at least one of 11 predefined skin locations,³³ including predilection sites for AD in infants, namely the cheeks and extensor surfaces of the extremities, as previously reported.^{5,8} The dry skin evaluation and severity is in line with the principles of the Dry skin/Ichtyosis and Severity Index (DASI), but without their score of erythema.³⁴ Dry skin severity is categorized into mild, moderate or severe dry skin.

Filaggrin analysis

DNA isolated from blood was genotyped using a TaqManbased allelic discrimination assay (Applied Biosystems, Foster City, CA, USA), as previously described.³² We defined *FLG* mutations ('mutation yes') as being a carrier of any of the following mutations in the *FLG* gene: R501X, 2282del4 and R2447X, the most common loss-of-function mutations in the European population.³²

Exposures

- Skin intervention (SI) includes the skin intervention group and the combined interventions group.
- No skin intervention (NSI) includes the control group and the food intervention group.
- Being an FLG mutations carrier was defined as being a carrier of any of the three mutations: R501X, 2282del4 and R2447X.³²

Outcomes

- TEWL through infancy, assessed at 3, 6 and 12 months of age, and for sensitivity analysis at each time-point.
- Presence of dry skin through infancy, and for sensitivity analysis, dry skin (1) at any or multiple time-points, (2) as moderate/severe at any time-point and (3) on cheeks and/ or extensor surfaces of the extremities at any time-point.

Statistical analysis

Categorical variables are presented as numbers and percentages. Continuous variables are presented as means, SDs and minimum (min) – maximum (max).

The effect of the skin intervention was analysed based on a modified intention-to-treat principle, excluding individuals where the outcome was not assessed. The average effect of the intervention on TEWL (continuous outcome) and dry skin (binary outcome) over time, from 3 to 12 months of age, was assessed using linear and logistic mixed-effects modelling, with the intervention as fixed effect, and participant ID (random intercept) and time-point (random slope) as random effects. Simple linear and logistic regression was used to analyse TEWL and the risk of dry skin, respectively, at each time-point separately. Potential effects of carrying FLG mutations on associations between skin intervention and TEWL and dry skin, respectively, were analysed by adjusting for FLG mutation status and including an interaction term (between skin intervention and FLG mutation status) in the mixed-effects models, with further testing by a three-way-interaction analysis (skin intervention*FLG mutation status*time-point) for the primary outcome (Table S1; see Supporting Information).

No background characteristics were included as potential confounders in adjusted models, as the random assignment to the SI or NSI group provided balanced baseline data in the two groups.²⁷

To account for missing data in the primary outcome variables, a sensitivity analysis was performed for TEWL in the study population using multiple imputation with multivariate normal distribution with an upper limit of 20 imputations at each time-point.

Statistical analysis was performed using Stata/MP v.16 software (StataCorp LLC, College Station, TX, USA), and *P*-values were considered significant if less than 0.05.

Results

The 2153 participants were recruited in Norway (79%) and Sweden (21%), with 53% boys. No significant differences in

background characteristics were observed between the 995 infants in the SI and 1158 infants in the NSI groups (Table 1), while they were generally similar in the 241 infants with missing information on TEWL, except for study location, living environment, maternal age and educational level of the caregivers (Table S2; see Supporting Information).

Skin intervention and transepidermal water loss

Infants randomized to the SI group had a significantly higher average TEWL [95% confidence interval (CI)] of 0.4 g m⁻² h⁻¹ (0.1–0.7, P=0.004) from 3 to 12 months of age (Figure 2a) with an intra-cluster correlation of 0.20 (95% CI 0.16–0.25).

The effect was dominated by significantly higher TEWL (95% CI) at 3 months in infants with SI compared with infants in the NSI group [8.6 (8.3–9.0) vs. 7.6 (7.3–7.9); P < 0.001], while TEWL was similar in the two groups at 6 and 12 months of age (Table 2). A dose–response was identified for TEWL at 3 months in relation to degree of adherence to the skin intervention (Figure 2b and Table S3; see Supporting Information). The finding of increased TEWL attributed to the skin intervention was supported by analysis of all four intervention groups (Figure S1; see Supporting Information).

In the analysis of multiple imputation of missing TEWL variables, the TEWL in the NSI and SI groups was

 Table 1
 Distribution in background characteristics, in the study population in total and according to allocation to the Skin intervention (SI) or No skin intervention (NSI) groups

	All infants <i>N</i> =2153ª <i>n</i> (%)	NSI N=1158ª n (%)	SI N=995° n (%)	<i>P</i> -value ^b
Sex of infant				0.545
Bov	1136 (52.8)	618 (53.4)	518 (52.1)	
Study location				0.367
Oslo (Norway)	1401 (65.1)	768 (66.3)	633 (63.6)	
Østfold (Norway)	292 (13.6)	155 (13 4)	137 (13 8)	
Stockholm (Sweden)	460 (21.4)	235 (20.3)	225 (22.6)	
Living environment ^a				0.038
City, densely populated	780 (39.7)	422 (40.0)	358 (39.5)	
City less densely populated	737 (375)	380 (36 0)	357 (39 4)	
Suburb	311 (15.8)	165 (15 6)	146 (16 1)	
Village	38 (1.9)	27 (2 6)	11 (1 2)	
Countryside	97 (4 9)	62 (5.9)	35 (3.9)	
Delivery mode	07 (110)	02 (0.0)		0 450
C-section ves	348 (16 2)	181 (15 6)	167 (16 8)	01100
Gestational age at birth days	0.10(1012)			0.530
Mean (SD)	280 4 (9 5)	280 5 (9 6)	280 3 (9 3)	0.000
Birth weight	200.1 (0.0)	200.0 (0.0)	200.0 (0.0)	0.983
Mean (SD)	3570 6 (480 1)	3570 4 (474 2)	3570 8 (4871)	0.000
ELC mutation ($M_{\rm c}$ 1692)		007011(17112)		0.430
V_{00}	152 (0, 1)	79 (9 6)	75 (0 7)	0.100
Natarnal aga	155 (9.1)	78 (8.0)	75 (9.7)	0 220
Moon (SD)	22 5 (4 1)	22.6(4.1)	22 4 (4 2)	0.320
Paternal ago	32.3 (4.1)	32.0 (4.1)	32.4 (4.2)	0 501
Mean (SD)	34.8 (5.4)	34 8 (5 4)	34 7 (5 5)	0.531
Educational loval mother ^a	34.0 (3.4)	34.0 (3.4)	34.7 (3.3)	0 740
Primary school	12 (0.6)	6 (0,6)	6 (0 7)	0.743
High school	188 (9.6)	100 (9.5)	88 (9 7)	
	608 (31 1)	326 (31.0)	282 (31.2)	
Higner education < 4 years		520 (51.0)		
Higher education \geq 4 years	1089 (55.7)	590 (56.2)	499 (55.2)	
PhD	55 (2.8)	28 (2.7)	27 (3.0)	
Other	2 (0.1)	0 (0.0)	2 (0.2)	
Educational level, partner ^a				0.680
Primary school	21 (1.1)	13 (1.3)	8 (0.9)	
High school	339 (18.0)	174 (17.2)	165 (18.8)	
Higher education $<$ 4 years	578 (30.6)	312 (30.9)	266 (30.4)	
Higher education > 4 years	865 (45.8)	471 (46.6)	394 (45.0)	
PhD	63 (3.3)	29 (2.9)	34 (3.9)	
Other	21 (1.1)	12 (1.2)	9 (1.0)	
Firstborn child of mother				0.051
Yes	1289 (60.0)	671 (58.0)	618 (62.2)	
Parental atopic diseaseª,c				
Maternal	808 (41.2)	453 (42.9)	355 (39.1)	0.092
Paternal	688 (35.0)	375 (35.5)	313 (34.3)	0.581
Either parent	1226 (63.9)	674 (65.4)	552 (62.2)	0.136
1				

No skin intervention, no intervention group + food intervention group; Skin intervention, skin intervention group + combined intervention group.^aDifferent denominators are due to missing data. ^b*P*-values for χ^2 tests and independent samples *t*-tests for differences in distribution when comparing children with and without the skin intervention. ^cDoctor diagnosed any of asthma, atopic dermatitis, allergic rhinitis or food allergy at enrolment (mother) or 34 weeks (father).



Figure 2 (a) TEWL, in g m⁻² h⁻¹ (95% CI), in the 2153 infants in the Skin intervention and No skin intervention groups, based on the linear mixed-effects model with random effects on the individual for repeated measurements at 3, 6 and 12 months of age. The figure was done using Stata/MP v.16 software. (b) TEWL (95% CI) at 3 months of age in the infants with Skin intervention (N=995), according to degree of adherence to the skin intervention. Partial adherence is based on oil baths only.CI, confidence interval; TEWL, transepidermal water loss

similar to the primary analysis (Table S4; see Supporting Information).

Skin intervention and dry skin

On average over time, from 3 to 12 months of age, the infants in the SI group were significantly less likely to have dry skin compared with the infants in the NSI group [odds ratio (OR) 0.71, 95% CI 0.64-0.79]. Dry skin was observed in 82% (1771 of 2153) of the infants at any of the three timepoints, ranging from 56% (1172 of 2112) at 3 months, to 58% (1171/2012) and 57% (1087/1899) at 6 and 12 months, respectively, with significant differences related to the skin intervention at 3 and 6 months (Table 2). The significantly decreased odds of dry skin at 3 months (OR 0.73, 95% CI 0.61-0.86) and 6 months (OR 0.68, 95% CI 0.57-0.81), as well as at all three time-points (OR 0.66, 95% CI 0.54-0.82), by the skin intervention are presented in detail in Table S5 (see Supporting Information). The skin intervention was not significantly associated with the presence of moderate/severe dry skin through infancy, as shown in Table S6

(see Supporting Information). However, infants randomized to the skin intervention had dry skin observed significantly less often on the cheeks and/or extensors at all three timepoints (Table S6).

Skin intervention, transepidermal water loss and dry skin in *FLG* mutation carriers

No interaction was observed between the skin intervention and *FLG* mutation status (interaction term *P*-values 0.505 and 0.211 in the adjusted mixed-effects models with TEWL and dry skin, respectively). Irrespective of the intervention, the 153 infants carrying *FLG* mutations showed significantly higher TEWL at all three time-points compared with the 1530 infants without an *FLG* mutation (Table S7; see Supporting Information). Being an *FLG* mutation carrier did not increase TEWL further when subjected to the skin intervention (Table 2). At 3 months, TEWL in infants in the SI group without *FLG* mutations was similar to infants in the NSI group with a *FLG* mutation (Table 2 and Figure 3).

Discussion

In approximately 2000 Scandinavian infants from the PreventADALL study, the TEWL was significantly higher in 3-month-old infants subjected to skin intervention with frequent mineral-based oil baths from 2 weeks of age through 8 months, compared with those without skin intervention. The increase in TEWL was dose-dependent according to adherence to the intervention. At 6 and 12 months the TEWL no longer differed between the intervention groups. Dry skin was significantly less common in infants in the SI group compared with the NSI group. At 3 months, the infants with *FLG* mutations had a similar TEWL as the infants in the SI group.

To our knowledge this is the first RCT that demonstrates an increased TEWL at 3 months in infants subjected to a skin intervention consisting of paraffinum liquidum bath oil and face cream. Our finding is partly supported by a substudy from the Irish observational birth cohort BASELINE study of 1500 children, showing that TEWL at 2 months of age was significantly higher in infants with frequent emollient baths and leave-on emollients from 2 days of age compared with three groups with less treatment.²⁶ In the BASELINE study, the group with frequent bath emollient and infrequent leave-on emollient did not have a significantly higher TEWL compared with the infrequent bath/leave-on emollient group. Because the skin intervention in the present study was bath emollient at least 4 days per week, and as we also found a dose-response effect on TEWL, the amount of emollient used per week might be important for skin barrier function. Although the BASELINE study measured TEWL at 2, 6, 12 and 24 months, only TEWL at 2 months was reported.²⁶ In contrast to our findings, a German study with 60 individuals aged 0.5-85 (mean 40) years with clinically dry skin and similar TEWL at enrolment observed a significant decrease in TEWL after 28 days of bathing with 1 mL of soy bean oil per 5 L of water every second day.²⁰ In a review by Wilborn et al., inconsistent TEWL results were found in 15 of 42 studies with different skincare regimes in neonates and infants.²³ In partial support of our findings, a study including 115 British healthy newborns randomized

	No skin intervention <i>N</i> =1158	Skin intervention <i>N</i> =995	<i>P</i> -value ^a
3 months of age			
TEWL (95% CI; SD)	7.62 (7.31–7.93; 5.02)	8.64 (8.29-8.99; 5.23)	< 0.001
FLG yes	8.89 (7.53–10.26; 5.63)	8.95 (7.72–10.19; 4.91)	0.950
<i>FLG</i> no	7.31 (6.97–7.66; 4.76)	8.50 (8.09-8.91; 5.10)	< 0.001
Dry skin, <i>n</i> (%)	671/1135 (59.1)	501/977 (51.3)	< 0.001
FLG yes + dry skin yes	58/887 (6.5)	56/758 (7.4)	0.499
FLG ves + drv skin no	18/887 (2.0)	17/758 (2.2)	0.765
FLG no + drv skin ves	480/887 (54.1)	345/758 (45.5)	0.001
FLG no + dry skin no	331/887 (37.3)	340/758 (44.9)	0.002
6 months of age			
TEWL (95% CI; SD)	8.17 (7.91–8.44; 4.28)	8.43 (8.11–8.75; 4.75)	0.217
FLG yes	9.45 (8.09–10.82; 5.68)	9.94 (8.45–11.43; 5.86)	0.627
<i>FLG</i> no	7.98 (7.68–8.29; 4.16)	8.22 (7.86–8.58; 4.53)	0.321
Dry skin, <i>n</i> (%)	678/1084 (62.6)	493/928 (53.1)	< 0.001
FLG yes+dry skin yes	64/853 (7.5)	47/730 (6.4)	0.408
<i>FLG</i> yes + dry skin no	9/853 (1.1)	19/730 (2.6)	0.020
FLG no + dry skin yes	484/853 (56.7)	355/730 (48.6)	0.001
FLG no + dry skin no	296/853 (34.7)	309/730 (42.3)	0.002
12 months of age			
TEWL (95% CI; SD)	6.90 (6.66–7.14; 3.33)	6.72 (6.47–6.96; 3.26)	0.293
FLG yes	7.68 (6.56–8.80; 4.07)	7.22 (6.28–8.15; 3.18)	0.534
<i>FLG</i> no	6.76 (6.49–7.03; 3.22)	6.72 (6.41–7.02; 3.38)	0.843
Dry skin, <i>n</i> (%)	607/1033 (58.8)	480/866 (55.4)	0.144
<i>FLG</i> yes+dry skin yes	56/823 (6.8)	42/687 (6.1)	0.587
<i>FLG</i> yes + dry skin no	11/823 (1.3)	22/687 (3.2)	0.014
<i>FLG</i> no+dry skin yes	447/823 (54.3)	349/687 (50.8)	0.173
<i>FLG</i> no+dry skin no	309/823 (37.5)	274/687 (39.9)	0.353

Table 2 Transepidermal water loss (TEWL), in g m⁻² h⁻¹ (95% confidence interval, CI), and prevalence of dry skin in the 2153 infants, at 3, 6 and 12 months of age, in Skin intervention and No skin intervention groups, respectively, as well as in infants with or without the presence of *FLG* mutations

No skin intervention, no intervention group + food intervention group; Skin intervention, skin intervention group + combined intervention group.^a*P*-values for χ^2 tests and independent samples *t*-tests.



Figure 3 Transepidermal water loss (TEWL), in g m⁻² h⁻¹ (95% CI), in 1683 infants with and without *FLG* mutations in the No skin intervention and Skin intervention groups. The figure was done using Stata/MP v.16 software.

to direct application of either sunflower oil, olive oil or no intervention found similar TEWL between the groups at 4 weeks. However, in both oil groups the organization of lipids throughout the stratum corneum was significantly impaired, which is associated with decreased skin barrier function.³⁵

Our finding of reduced skin barrier function at 3 months in the skin intervention group is also supported by studies reporting increased TEWL from leave-on emollients;^{18,23,24} however, it is difficult to directly compare studies using different emollients, as they can have varying effects on the skin barrier due to their diverse compositions. A substudy from the British EAT cohort observed that increasing the moisturizing frequency in 1221 infants at 3 months of age was associated with higher concurrent TEWL,²⁴ and half of the emollients used were oils, olive oil being the most common (20%), followed by oils containing paraffinum liquidum (10%). A German pilot study, including 50 infants at high risk for AD, found no difference in TEWL at 6 and 12 months between the intervention group applying daily emollient containing a prebiotic *Vitreocilla filiformis* lysate and the controls.²⁵

In contrast to our findings, a Japanese study including 227 neonates comparing moisturizing skincare (bathing every 2 days and using lotion daily) to daily bathing without lotion (controls), found a significantly lower face TEWL at 3 months in the intervention group compared with the controls.²² A German study including 64 neonates in four similar-sized groups with the intervention done twice-weekly (bathing with wash gel, bathing and cream, bathing with wash gel plus cream, and bathing with water) found significantly lower TEWL at 8 weeks of age in the intervention groups with cream compared with the group bathing with water only.³⁶ In both the above-mentioned studies, the bathing frequency in the intervention groups was considerably lower than in our study.

Our finding that infants subjected to the skin intervention had lower prevalence of dry skin than the control group is in line with the Norwegian pilot study³⁷ preceding the PreventADALL study,²⁷ using a similar skin intervention. The skin intervention from 6 weeks of age reduced xerosis at 6 months in the 24 infants having regular mineral-based oil baths compared with the 32 controls.³⁷ Some of the studies in the review by Wilborn et al.23 included dry skin as one of their outcomes, and in contrast to our findings most of these studies did not find any difference in dry skin prevalence between the intervention groups; however, all these studies had different designs, different interventions and much shorter follow-up periods compared with our study.²³ The finding that the skin intervention in our study reduced the appearance of dry skin on the cheeks and/or extensor surfaces of the extremities is novel, and somewhat contradictory to our previous findings of increased TEWL among infants with dry skin, especially on the cheeks and extensors.⁵ It is peculiar that the oil bath intervention seems to improve the appearance of dry skin, thereby masking the reduced skin barrier function seen as increased TEWL.

There are many studies on the effect of emollients on the dysfunctional skin barrier presenting as eczema lesions and dry skin seen in patients with AD.^{16,17,19} The majority of these show reduction in AD severity and reduction of flares when leave-on emollients are used, ^{16,17} but dry skin was not included as an outcome. In the British BATHE study, including 482 children between 1 and 11 years with AD, the intervention group using emollient bath additives regularly for 12 months did not identify any difference in AD severity,¹⁹ in support of our findings. A review by Lodén from 2012, including 11 studies investigating the effect of leave-on moisturizers on clinically dry skin, found a positive effect on dryness in all the studies, but all the moisturizers contained humectants such as urea or ammonium lactate.²¹

To the best of our knowledge, our finding of similar TEWL at 3 months between infants with or without *FLG* mutations subjected to frequent oil baths in a general unselected infant population is novel. While TEWL in infants with *FLG* mutation was not affected when subjected to frequent oil baths, it was significantly increased in infants without *FLG* mutations and reached the same level as for mutation carriers not subjected to the skin intervention. A British study by Flohr *et al.* found higher TEWL in *FLG* mutation carriers at 3 months of age, but the study did not include any emollient treatments.³⁸

Regarding clinical implications, the findings in the present study support that regular mineral-based emollients as a preventive measure do not improve the skin barrier function, which may be one of the reasons why the skin intervention in the PreventADALL study did not prevent AD. We have demonstrated that mineral oil bathing increases TEWL in a dose-response pattern at 3 months and thus may transiently even worsen the skin barrier function. The impaired skin barrier may facilitate penetration of allergens, as demonstrated in the BEEP study where emollients increased the risk of allergic sensitization,²⁸ and in the EAT study, where increasing use of emollients, oils in particular, increased the risk of food allergy.²⁴ However, we need to highlight that the amount of mineral oil used in our intervention was considerably higher than other oil bath intervention studies^{20,23,26} where an increase in TEWL was not found. It can be hypothesized that at considerably lower oil dosage, the skin barrier

may not be affected. Based on our results, we cannot advise against adding a few mL of mineral oil to the bath water, but we may suggest that frequent oil bathing of infants is not advisable. And there is certainly no evidence of a positive effect. It is also important to keep in mind that in the first few months of life the skin goes through a physiological maturation and there is no evidence that any skincare product, including bath oil, may be beneficial for the maturation process of healthy skin.²³

The strengths of our study are: a large, prospective RCT from a general population; high follow-up rates (89.1%, 84.4% and 79.7% at the 3-, 6- and 12-month investigations, respectively); stringent skin assessment by trained personnel as well as standardized TEWL measurements in most of the participants (79.3%, 77.7%, and 58.7%, at 3, 6 and 12 months, respectively); and close monitoring of adherence to the skin intervention.²⁷ Most of the study participants originate from the Nordic countries, limiting the generalizability. Another considerable limitation is the low full protocol adherence of 32% for the skin intervention. However, 50% and 58% were able to perform oil baths for an average of at least 3.5 days per week, or 2.5 days per week, respectively.

In conclusion, in the general population-based PreventADALL study, the skin intervention comprising of mineral-based bath oil several times per week transiently increased TEWL at 3 months of age in a dose-dependent manner independent of *FLG* mutation status. Surprisingly, the appearance of any dry skin was less common in the infants subjected to the skin intervention compared with the infants with no skin intervention, especially at 3 and 6 months of age. The findings of the skin intervention leading to reduced skin barrier function may partly explain why AD by the age of 12 months was not prevented by the skin intervention.²⁷

Acknowledgements

We sincerely thank all families participating in the Preventing Atopic Dermatitis and ALLergies (PreventADALL) study and the study personnel contributing to enrolling and managing the study: H. Aaneland, S. Asad, A. Asarnoj, K.E.S. Bains, A. Berglind, J. Björk, O.C. Lødrup Carlsen, Å. Wik Despriée, K.M.A. Endre, T. Aspelund Fatnes, S. Götberg, P.A. Granlund, M. Gudbrandsgard, H.K. Gudmundsdóttir, G. Haugen, M. Rønning Kjendsli, I. Kreyberg, L. Landrø, A. Lie, C.A. Olsson Mägi, N. Nilsson, M. Nordenbrand, C.M. Saunders, K. Sedergren, N. Sedergren, Sigrid Sjelmo, P. Söderman, L.J. Sørdal, S. Ganrud Tedner, E. Tegnerud, M.R. Værnesbranden and J. Wiik, and in memoriam K.H. Carlsen.

Funding sources

The PreventADALL study was supported by several public and private funding bodies with no influence on design, conduct or analyses. The PreventADALL study has received funding from the following sources: The Regional Health Board South East, The Norwegian Research Council, Oslo University Hospital, The University of Oslo, Health and Rehabilitation Norway, The Foundation for Healthcare and Allergy Research in Sweden – Vårdalstiftelsen, The Swedish Asthma and Allergy Association's Research Foundation, The Swedish Research Council – the Initiative for Clinical Therapy Research, The Swedish Heart-Lung Foundation, SFO-V Karolinska Institutet, Østfold Hospital Trust, by unrestricted grants from the Norwegian Association of Asthma and Allergy, The Kloster Foundation, Thermo-Fisher, Uppsala, Sweden (through supplying allergen reagents) and Fürst Medical Laboratory, Oslo, Norway (through performing IgE analyses). Norwegian Society of Dermatology and Venerology, Arne E. Ingels Bequest, Region Stockholm (ALF-project and individual grants), Forte, Swedish Order of Freemasons Foundation Barnhuset, The Sven Jerring Foundation, The Hesselman Foundation, The Magnus Bergwall Foundation. The Konsul Th C Bergh's Foundation. The Swedish Society of Medicine, The King Gustaf V 80th Birthday Foundation, Karolinska Institutet grants, The Cancer and Allergy Foundation, The Pediatric Research Foundation at Astrid Lindgren Children's Hospital, The Samaritan Foundation for Pediatric Research, Barnestiftelsen at Oslo University Hospital, Roche, The Frithjof Nansen Institute.

Conflicts of interest

E.M.R. has received honoraria for lectures from Leo Pharma, Novartis, Norwegian Asthma and Allergy Association, Norwegian Psoriasis and Eczema Association, and Sanofi Genzyme. K.C.L.C. reports that her institution has received an honorarium from Thermo Fisher Scientific for symposium presentation. C.M.J. has received honoraria for lectures from the Norwegian Medical Association outside the submitted work and M.L. reports personal fees from MSD. All other authors declare no conflicts of interest.

Data availability

The data underlying this article cannot be shared publicly because the study is still ongoing and due to the privacy of individuals that participated in the study. The data may be shared on reasonable request to the corresponding author.

Ethics statement

The study was conducted according to the guidelines of the Declaration of Helsinki, and was approved by the Regional Committee for Medical and Health Research Ethics in Norway (2014/518) and the Swedish Ethical Review Authority in Sweden (2014/2242-31/4 and 2018/1437-32), and registered at https://www.clinicaltrials.gov, NCT02449850 (https://clinicaltrials.gov/study/NCT02449850). Informed written consent was obtained from all mothers at enrolment, and from parent(s) at newborn inclusion.

Supporting Information

Additional Supporting Information may be found in the online version of this article at the publisher's website.

References

- Hadi HA, Tarmizi AI, Khalid KA *et al.* The epidemiology and global burden of atopic dermatitis: a narrative review. *Life (Basel)* 2021; 11:936. https://doi.org/10.3390/life11090936.
- 2 Wollenberg A, Werfel T, Ring J *et al.* Atopic dermatitis in children and adults – diagnosis and treatment. *Dtsch Arztebl Int* 2023; **120**:224–34.

- 3 Seidenari S, Giusti G. Objective assessment of the skin of children affected by atopic dermatitis: a study of pH, capacitance and TEWL in eczematous and clinically uninvolved skin. *Acta Derm Venereol* 1995; **75**:429–33.
- 4 De Marchi F, Piacentini GL, Piazza M *et al.* Correlation of skin barrier impairment in atopic dermatitis with aeroallergen sensitization. *Allergy Asthma Proc* 2015; **36**:e127–33.
- 5 Rehbinder EM, Winger AJ, Landrø L *et al.*; PreventADALL study group. Dry skin and skin barrier in early infancy. *Br J Dermatol* 2019; **181**:218–19. https://doi.org/10.1111/bjd.17626.
- 6 Huppop F, Dahnhardt-Pfeiffer S, Folster-Holst R. Characterization of classical flexural and nummular forms of atopic dermatitis in childhood with regard to anamnestic, clinical and epidermal barrier aspects. *Acta Derm Venereol* 2022; **102**:adv00664.
- 7 Bohme M, Svensson A, Kull I, Wahlgren CF. Hanifin's and Rajka's minor criteria for atopic dermatitis: which do 2-year-olds exhibit? *J Am Acad Dermatol* 2000; **43** (5 Pt 1):785–92.
- 8 Rehbinder EM, Advocaat Endre KM, Lodrup Carlsen KC et al. Predicting skin barrier dysfunction and atopic dermatitis in early infancy. J Allergy Clin Immunol Pract 2020; 8:664–73 e5.
- 9 Berents TL, Lødrup Carlsen KC, Mowinckel P et al. Transepidermal water loss in infancy associated with atopic eczema at 2 years of age: a population-based cohort study. Br J Dermatol 2017; 177:e35–7. https://doi.org/10.1111/bjd.15157.
- 10 Gupta J, Grube E, Ericksen MB *et al.* Intrinsically defective skin barrier function in children with atopic dermatitis correlates with disease severity. *J Allergy Clin Immunol* 2008; **121**:725–30 e2.
- 11 Wang S, Shen C, Zhao M *et al.* Either transepidermal water loss rates or stratum corneum hydration levels can predict quality of life in children with atopic dermatitis. *Pediatr Investig* 2021; 5:277–80.
- 12 Levin J, Friedlander SF, Del Rosso JQ. Atopic dermatitis and the stratum corneum: part 1: the role of filaggrin in the stratum corneum barrier and atopic skin. *J Clin Aesthet Dermatol* 2013; 6:16–22.
- 13 Bager P, Wohlfahrt J, Thyssen JP, Melbye M. Filaggrin genotype and skin diseases independent of atopic dermatitis in childhood. *Pediatr Allergy Immunol* 2016; 27:162–8.
- 14 Brown SJ, Irvine AD. Atopic eczema and the filaggrin story. *Sem Cutan Med Surg* 2008; **27**:128–37.
- 15 Thyssen JP, Bikle DD, Elias PM. Evidence that loss-of-function filaggrin gene mutations evolved in Northern Europeans to favor intracutaneous vitamin D3 production. *Evol Biol* 2014; **41**:388–96.
- 16 van Zuuren EJ, Fedorowicz Z, Arents BWM. Emollients and moisturizers for eczema: abridged Cochrane systematic review including GRADE assessments. Br J Dermatol 2017; **177**:1256–71.
- 17 Ridd MJ, Santer M, MacNeill SJ *et al.* Effectiveness and safety of lotion, cream, gel, and ointment emollients for childhood eczema: a pragmatic, randomised, phase 4, superiority trial. *Lancet Child Adolesc Health* 2022; **6**:522–32.
- 18 Kelleher MM, Phillips R, Brown SJ et al. Skin care interventions in infants for preventing eczema and food allergy. Cochrane Database Syst Rev 2022; 11:CD013534.
- 19 Santer M, Rumsby K, Ridd MJ *et al.* Adding emollient bath additives to standard eczema management for children with eczema: the BATHE RCT. *Health Technol Assess* 2018; 22:1–116. https:// doi.org/10.3310/hta22570.
- 20 Kottner J, Kanti V, Dobos G *et al.* The effectiveness of using a bath oil to reduce signs of dry skin: a randomized controlled pragmatic study. *Int J Nurs Stud* 2017; 65:17–24.
- 21 Lodén M. Effect of moisturizers on epidermal barrier function. *Clin Dermatol* 2012; **30**:286–96. https://doi.org/10.1016/j.clindermatol.2011.08.015.
- 22 Yonezawa K, Haruna M, Matsuzaki M *et al.* Effects of moisturizing skincare on skin barrier function and the prevention of skin problems in 3-month-old infants: a randomized controlled trial. *J Dermatol* 2018; **45**:24–30.

- 23 Wilborn D, Amin R, Kottner J, Blume-Peytavi U. Skin care in neonates and infants: a scoping review. *Skin Pharmacol Physiol* 2023; **36**:51–66.
- 24 Perkin MR, Logan K, Marrs T *et al.* Association of frequent moisturizer use in early infancy with the development of food allergy. *J Allergy Clin Immunol* 2021; **147**:967–76.e1.
- 25 Harder I, Stolzl D, Sander N et al. Effects of early emollient use in children at high risk of atopic dermatitis: a German pilot study. Acta Derm Venereol 2023; 103:adv5671.
- 26 O'Connor C, Livingstone V, Hourihane JOB *et al.* Early emollient bathing is associated with subsequent atopic dermatitis in an unselected birth cohort study. *Pediatr Allergy Immunol* 2023; 34:e13998. https://doi.org/10.1111/pai.13998.
- 27 Skjerven HO, Rehbinder EM, Vettukattil R et al. Skin emollient and early complementary feeding to prevent infant atopic dermatitis (PreventADALL): a factorial, multicentre, cluster-randomised trial. Lancet 2020; 395:951–61.
- 28 Chalmers JR, Haines RH, Bradshaw LE *et al.* Daily emollient during infancy for prevention of eczema: the BEEP randomised controlled trial. *Lancet* 2020; **395**:962–72.
- 29 Lødrup Carlsen KC, Rehbinder EM, Skjerven HO *et al.*; study group. Preventing Atopic Dermatitis and ALLergies in children – the PreventADALL study. *Allergy* 2018; **73**:2063–70. https://doi. org/10.1111/all.13468.
- 30 Skjerven HO, Lie A, Vettukattil R *et al.* Early food intervention and skin emollients to prevent food allergy in young children (PreventADALL): a factorial, multicentre, cluster-randomised trial. *Lancet* 2022; **399**:2398–411.

- 31 Rogiers V, Group E. EEMCO guidance for the assessment of transepidermal water loss in cosmetic sciences. *Skin Pharmacol Appl Skin Physiol* 2001; **14**:117–28.
- 32 Hoyer A, Rehbinder EM, Färdig M *et al.* Filaggrin mutations in relation to skin barrier and atopic dermatitis in early infancy. *Br J Dermatol* 2022; **186**:544–52.
- 33 Carson CG, Rasmussen MA, Thyssen JP *et al.* Clinical presentation of atopic dermatitis by filaggrin gene mutation status during the first 7 years of life in a prospective cohort study. *PloS One* 2012; **7**:e48678.
- 34 Serup J. EEMCO guidance for the assessment of dry skin (xerosis) and ichthyosis: clinical scoring systems. *Skin Res Technol* 1995; **1**:109–14.
- 35 Cooke A, Cork MJ, Victor S *et al.* Olive oil, sunflower oil or no oil for baby dry skin or massage: a pilot, assessor-blinded, randomized controlled trial (the Oil in Baby SkincaRE [OBSeRvE] Study). Acta Derm Venereol 2016; **96**:323–30.
- 36 Garcia Bartels N, Scheufele R, Prosch F et al. Effect of standardized skin care regimens on neonatal skin barrier function in different body areas. *Pediatric Dermatol* 2010; 27:1–8.
- 37 Kvenshagen BK, Carlsen KH, Mowinckel P et al. Can early skin care normalise dry skin and possibly prevent atopic eczema? A pilot study in young infants. Allergol Immunopathol (Madr) 2014; 42:539–43. https://doi.org/10.1016/j.aller.2014.06.003.
- 38 Flohr C, England K, Radulovic S *et al.* Filaggrin loss-of-function mutations are associated with early-onset eczema, eczema severity and transepidermal water loss at 3 months of age. *Br J Dermatol* 2010; **163**:1333–6.